Author's response to reviews

Title: The role of Traditional Birth Attendants in urban and rural Tanzania: a mixed-methods study

Authors:

Constanze D Pfeiffer (constanze.pfeiffer@unibas.ch)
Rosemarie Mwaipopo (ny_lila@yahoo.com)

Version: 7 Date: 5 February 2013

Author's response to reviews: see over
Re: MS: 1249840671809233
The role of Traditional Birth Attendants in urban and rural Tanzania: a mixed-methods study
Constanze D Pfeiffer and Rosemarie Mwaipopo

Dear Dr. Crow,

We thank you again for accepting the manuscript in principle. With the support of a colleague we have now completely copyedited our manuscripts. We would be delighted if the editors would consider the final version for publication in BMC Pregnancy and Childbirth.

Yours sincerely,

Dr. Constanze Pfeiffer